


Present: Alexei, Flemming, Gene, Mike, Tonko, Jim, Yuri, and myself



== Hardware

	 — recovered survey probe 

	 	 — silicon detector is currently out and obstructs some of the survey points

	 	 — proposes delaying TPC survey by ~2 weeks

	 — Alexei on vacation for a week next week

	 — Discussed how to remove TPC anode wire(s) for examination

	 	 — several possible approaches

	 	 — will package in clean room bag

	 	 — will get wires from inner part of sector and outer part of sector 





== Electronics (Flemming)

	 — Christian and Tim still working on fixing RDOs (~6 recovered just by reprogramming)

	 	 — still several that need to be removed to be repaired

	 — Most work being done on east side (where most problems were)

	 — moving on with VHDL of outer sectors for DAQ upgrade

	 — currently at 4.9 kHz (internal goal is 7 kHz)

	 — memory upgrade for old PCs is done

	 	 — spares from Richard have arrived and are in storage





== Software

	 — will update on dE/dx next week

	 	 — finished first pass

	 	 — trying to recover momentum dependency 

	 	 — will cross-check to make sure new calibration is an improvement

	 — Tommy running some software tools from Yuri

	 	 — will test new vs old cluster finder 

	 	 — Tonko sent interface for testing, needs feedback





== AOB

	 — Jim will give the STAR run report at user’s meeting

	 	 — asking for input material

	 	 — talk will be very run specific

	 	 — Flemming and Alexei mentioned information from recent retreat (Flemming’s talk)

	 	 — Please send further suggestions directly to Jim





